Author's response to reviews

Title: Chemoradiotherapy and Consolidation Chemotherapy with Cisplatin and 5-fluorouracil in Anal Squamous Cell Carcinoma: Long-term results in 31 patients

Authors:

Byoung Chul Cho (cbc1971@yumc.yonsei.ac.kr)
Joong Bae Ahn (vvswm513@yumc.yonsei.ac.kr)
Won-Na Suh (suhwn@yumc.yonsei.ac.kr)
Jinsil Seong (jsseong@yumc.yonsei.ac.kr)
Jae Kyung Roh (jkroh@yumc.yonsei.ac.kr)
Joo Hang Kim (kijhang@yumc.yonsei.ac.kr)
Hyun Cheol Chung (unchung8@yumc.yonsei.ac.kr)
Joo Hyuk Sohn (oncosohn@yumc.yonsei.ac.kr)
Nam Kyuk Kim (namkyuk@yumc.yonsei.ac.kr)

Version: 4 Date: 13 July 2007

Author's response to reviews:

12 July 2007

Dear Editor-in-Chief,

Please find enclosed our manuscript entitled "Chemoradiotherapy and Consolidation Chemotherapy with Cisplatin and 5-fluorouracil in Anal Squamous Cell Carcinoma: Long-term results in 31 patients" which we are submitting for consideration for publication as an original article in 'BMC Cancer'.

The manuscript has not been published and not under consideration for publication in any other journals. All authors have read and approved this manuscript. In consideration of reviewing our submission, the undersigned authors transfer, assign, or otherwise convey all copyright ownership to 'BMC Cancer' in the event the work is published. Thank you for considering this manuscript for publication in 'BMC Cancer'.

Sincerely yours,

Joong Bae Ahn, M.D, Ph.D.
Yonsei Cancer Center, Yonsei University College of Medicine, 134 Shinchon-Dong, Seodaemun-Gu, Seoul, Korea.
Tel: 82-2-2228-8134; Fax: 393-3652
E-mail: vvswm513@yumc.yonsei.ac.kr